\begin{table} %t1
%\centering
\par
\caption{Sources of $V$-, $K$-band and RV data.} 
\par
\begin{tabular}{rrrrl} 
\hline\hline 
Name       & $V$                               & $K$            & $RV$            &  Data not considered$^1$ \\  
\hline 
AQ Pup     & $1,2,3,4,5,84$ & $6,7,8$   & $1,10,11,44$     & $3,-,-$ JD $<$ 34~000, JD $>$ 49~700 \\
BB Sgr     & $2,3,16,63$    & $6,24$    & $20,33,39,44$    & $-,-,44$ \\
beta Dor   & $3,11,16$      & $6,18$    & $11,19,20,21,22$ & $-,-,20$     \\  
BF Oph     & $2,3,16,23,63$ & $6,18,24$ & $10,20,39,44,76$ & JD $<$ 36~000 \\
BG Lac     & $2,3,12,13$    & 12      & $14,15,44$ \\
BN Pup     & $1,3,5,16$     & $6,7$     & $1,17$ \\                                                  % RV
CD Cyg     & $2,3,64$       & 24      & $33,44$ \\
CV Mon     & $2,3,13,16,23$    & $6,24$  & $25,26,44$ \\
delta Cep  & $2,3,11,12,13,27$ & $12,98$    & $4,9,14,25,28,29$ & $-,-,9$ JD $<$ 43~500 \\              % K-mag
DT Cyg     & $2,3,4,27,30,31$  & $24,32$ & $4,9,28,29,33,34$ & $-,32,9+28$ JD $<$ 34~000 \\             % K-mag
eta Aql    & $2,3,12,13,16,27,35$ & $12,24$ & $9,14,20,25,28,29,36,37$ & $-,-,9$ \\
FF Aql     & $2,3,11,16,27,30,31$ & $18,24,32$ & $9,20,28,29,33,38$ & $3,32,9$ JD $<$ 44~000 \\         % K,  RV=binary
FM Aql     & $2,3,12,13,16,23$ & $12,24$ & $14,33,39,44$ & JD $<$ 30~000 \\
FN Aql     & $2,3,12,13,16,23$ & $12,24$ & $14,33,39$    & $-,-,39$ \\
KN Cen     & $1,3,5,11,16,40,84$ & 6 & $1,17,41$ & $-,-,41$ JD $<$ 40~000, JD $>$ 47~000 \\            % RV
KQ Sco     & $1,3,11$            & 6 & $1,42$ \\                                                   % RV
l Car      & $3,5,11,16,35$   & 6 & $11,19,20,23,43$ & $5,-,23$ JD $<$ 46~500 \\ 
LS Pup     & $1,3,11$         & 6 & 1 \\                                                         % RV
RS Pup     & $2,3,5,11,16,84$ & $6,8,24$ & $11,19,25,44$ & $-,-,44$ JD $<$ 42~000, JD $>$ 50~000 \\
RT Aur     & $2,12,27,31,45$  & $12,32$ & $9,28,29,33,37,46$ & $-,-,9 $ \\                             % K
RU Sct     & $2,3,16$         & 6     & $26,33,105$ &                   \\
RY Sco     & $1,2,3,5,11,16$  & 6     & $1,20,44$ \\
RY Vel     & $1,3,5,11,16$    & 6     & $1,11,17$ & JD $<$ 44~000, JD $>$ 50~000 \\
RZ Vel     & $1,3,5,11,16$    & 6 & $1,11,20,47$ \\
S Mus      & $3,11,23,40$     & $6,24$ & $10,11,20,22,41,47,48,49,50$ & $-,-,41$ JD $<$ 30~000 \\        % RV=binary
S Nor      & $3,11,23,40$     & $6,24$ & $4,10,20,41,47,49,51,52,53$  & $-,-,41$ JD $<$ 30~000 \\        % RV=binary
S Sge      & $2,3,11,12,13,23,27,30,40$ & $12,24,32$ & $9,14,28,29,33,37,54,55,56$ & $-,32,54$ \\       % RV=binary
SS Sct     & $2,3,16,23,63$   & 24 & $39,42,44,63$ & JD $<$ 30~000 \\                                % K
SU Cas     & $2,3,9,27,31$    & $12,32$ & $4,9,25,28,29,33,57,58,59,60,61,62$ & $-,32,9$ JD $<$ 43~000 \\  % RV=binary
SU Cyg     & $2,3,27,30,31$   & 24    & $9,28,33,106,107,108,109$           & $-,-,9 $ JD $<$ 35~000 \\  % RV=binary
S Vul      & $3,30$           & 6     & $33, 110$ & JD $<$ 45~400 \\
SV Vul     & $2,3,4,9,27,64$  & $6,12,24$ & $4,9,14,15,25,28,29,33,39,65$ & JD $<$ 45~500, JD $>$ 48600 \\ 
SW Vel     & $1,3,5,11,16,84$ & 6    & $1,11$ \\
SX Vel     & $3,23,40$        & 6    & 10   \\                                                    %RV
SZ Aql     & $2,3,11,12,16,23,64$ & $6,12,24$ & $11,14,23$ \\
SZ Tau     & $2,3,4,12,27,31$ & $6,12,32$ & $4,9,28,29,33$ & $-,32,9$ JD $<$ 43~000, JD $>$ 48~000\\
T Mon      & $1,2,3,11,16,35,64$  & $6,24,32$ & $4,9,25,29,33,37,48,65,66,67,68,69,70,71,72,73$ & $-,32,70$ JD $<$ 44~500 \\ %RV=binary
TT Aql     & $2,3,11,12,16,27,64,74$ & $12,24$ & $9,11,14,15,22,28,33,37,39,74$ & $-,-,74$ JD $<$ 44~000 \\
T Vel      & $3,16,75$        & $6,18$ & $10,20,75$ & $-,18,-$ \\
T Vul      & $2,3,12,27,31$   & $12,24,32,98$ & $4,9,14,28,29$ & $-,32,9$ JD $<$ 43~600\\
U Car      & $1,3,5,11,16,35$ & $6,24$ & $1,11,20$ \\
U Nor      & $1,5,11,16$      & 6    & $1,11,17,41$ \\
U Sgr      & $2,3,4,16,23,35,63$ & $6,24,32$ & $4,9,10,11,20,25,28,33,39,47,51,76,77,78$ & $-,32,9$ JD $<$ 37~000 \\
UU Mus     & $1,5,11,40,84$      & 6    & $1,11$ \\
U Vul      & $2,3,12,13,23,27$   & 12   & $4,14,33,37,44,79,80$ & $-,-,44$ \\                        % RV=binary
V340 Nor   & $3,4,82$             & 6     & $4,26$  \\                                             % RV
V350 Sgr   & $2,3,23,63,99$       & $18,24$ & $20,22,33,44,76,83,39$ & $-,24,-$ JD $<$ 42~000 \\        % RV=binary
V496 Aql   & $2,3,16,63$          & 24    & $9,10,20,33,76$ \\                                     % K, RV=binary
V Car      & $3,16,23$            & 6     & $10.20,47$ \\
V Cen      & $3,11,16,23,63$      & $6,24$  & $10,11,20,47,76$ & $-,24,-$ JD $<$ 42~000 \\
VW Cen     & $1,3,16,84$          & 6     & $1,41$  & $-,-,41$ \\                                    % RV
VY Car     & $1,3,5,11,16,99,100$ & $6,24$  & $1,11,20,41$ & JD $<$ 42~000 \\
VZ Cyg     & $2,3,12,31$          & $12,24$ & $4,14,33,39,44$ & JD $<$ 40~000 \\                      % RV=binary
VZ Pup     & $1,3,5,11,16$        & $6,7,8$ & $1,11$ \\
W Sgr      & $2,3,40,85$          & $24,32,86$ & $4,12,20,22,28,47,48,85,87,88,89$ \\                % K, RV=binary
WZ Car     & $1,3,5,11,16$        & 6     & $1,11$ \\
WZ Sgr     & $1,2,3,5,11,40$      & 6     & $1,11,33,39$ & JD $<$ 40~000 \\
X Cyg      & $2,3,27$             & $12,24,32,98$ & $4,14,25,29,33$ \\ 
X Lac      & $2,3,12,13$          & 12          & $4,14,33,44$ & $-,-,44$ \\
X Pup      & $2,3,5,11,16,23$     & 6           & $11,23,44$   & $5,-,-$ \\
X Sgr      & $2,3,11,16,35,99$    & $24,32,104$ & $9,10,11,20,28,47,52,101,102,103$ & $-,-,9$ JD $<$ 35~000 \\ % RV=binary
XX Cen     & $3,11,40,74$         & $6,24$  & $10,11,20,41,52,74$ & $-,24,74$ \\                       % RV=binary
Y Lac      & $2,3,12,31,90$       & 12    & $14,15,33,39,44$ &  JD $<$ 26~500 \\
Y Oph      & $1,2,3,16$           & $6,24$  & $1,9,19,20,28,33,34,91,92$ & $-,-,9$ JD $<$ 40~000 \\
\hline 
Y Sgr      & $2,3,11,16,23,35$    & $24,32$ & $9,11,19,20,28,46,49,93,94$ & $-,-,9$ JD $<$ 28~500 \\    % RV
YZ Sgr     & $2,3,23,40$          & 24    & $20,44$ \\
Z Lac      & $2,3,12,64$          & 12    & $14,33,39,44,80,95,96,97$ & JD $<$ 40~500  \\           % RV=binary
zeta Gem   & $2,3,4,27,35,64$     & $32,104$ & $4,9,19,28,29,33,37$ & $-,-,9$ \\                       % K
\hline 
\end{tabular}
\par
\smallskip
References:
\par
1~= Coulson \& Caldwell (\cite{Coulson1985a});   % SAAO Circulars, 9, 5
2~= Moffett \& Barnes (\cite{Moffett1984}); 
3~= Berdnikov et~al. (\cite{Berdnikov2000}), a datafile named ``cepheids-16-03-2006'' was retrieved from the ftp address listed in that paper;
4~= Bersier et~al. (\cite{Bersier1994}); 
5~= Madore (\cite{Madore1975}); 
6~= Laney \& Stobie (\cite{Laney1992}); 
7~= Schechter et~al. (\cite{Schechter1992});
8~= Welch (\cite{Welch1985});  
9~= Barnes et~al. (\cite{Barnes1987}), points with uncertainty flag ``:'' were removed; % ApJS, 65, 307
10~= Stibbs (\cite{Stibbs1955}); 
11~= Bersier (\cite{Bersier2002}), data points with weight 0 and 1 in the Geneva photometry were removed;  % ApJS, 140, 465
12~= Barnes et~al. (\cite{Barnes1997});  %PASP 109, 645
13~= Szabados (\cite{Szabados1980});   %Konkoly No.76  
14~= Barnes et~al. (2005); 
15~= Imbert (\cite{Imbert1999});
16~= Pel (\cite{Pel1976}); 
17~= Pont et~al. (\cite{Pont1994});
18~= Lloyd Evans (1980a); %SAAO Circulars, 1, 163
19~= Nardetto et~al. (\cite{Nardetto2006}); 
20~= Lloyd Evans (1980b);  %SAAO Circulars, 1, 257 
21~= Taylor \& Booth (\cite{Taylor1998}); 
22~= Petterson et~al. (\cite{Petterson2005}); 
23~= Caldwell et~al. (\cite{Caldwell2001});  %JAR 7,4
24~= Welch et~al. (\cite{Welch1984});  %ApJS, 54, 547
25~= Storm et~al. (2004); 
26~= Metzger et~al. (\cite{Metzger1992}); 
27~= Kiss (\cite{Kiss1998}); 
28~= Wilson et~al. (\cite{Wilson1989}); 
29~= Kiss (\cite{Kiss2000}); 
30~= Szabados (\cite{Szabados1991});   % Commun. Konkoly. , No.96 
31~= Szabados (\cite{Szabados1977});  %  Mitt. Sternw. ung. Akad. Wiss., No.70 
32~= Wisniewski \& Johnson (\cite{Wisniewski1968}); 
33~= Gorynya et~al. (\cite{Gorynya1998}, VizieR On-line Data Catalog: III/229); 
34~= Sanford (\cite{Sanford1935}); 
35~= Shobbrook (\cite{Shobbrook1992}); 
36~= Jacobsen \& Wallerstein (1981); 
37~= Evans (1976); 
38~= Evans et~al. (1990);  
39~= Joy (1937); 
40~= Walraven et~al. (1964); 
41~= Grayzeck (1978); 
42~= This paper, Table~\ref{TAB-DATA};
43~= Taylor et~al. (1997); 
44~= Barnes et~al. (1988), points with uncertainty flag ``:'' were removed;  % ApJS 66, 43
45~= Turner et~al. (2007); 
46~= Duncan (1908); 
47~= Lloyd Evans (1968);  % MN 168, 109
48~= Petterson et~al. (2004); 
49~= Campbell \& Moore (1928); 
50~= B\"ohm-Vitense et~al. (1990); 
51~= Mermilliod et~al. (1987);
52~= Feast (1967); 
53~= Breger (1970); 
54~= Evans et~al. (1993);
55~= Herbig \& Moore (1952);
56~= Breitfellner \& Gillet (1993);
57~= Adams \& Shapley (1918); 
58~= Abt (1959);
59~= Niva \& Schmidt (1979);
60~= Gieren (1976);
61~= H\"aupl (1988);
62~= Beavers \& Eitter (1986);
63~= Gieren (1981b);
64~= Szabados (1981);  %Comm 77
65~= Sanford (1956);
66~= Evans et~al. (1999); 
67~= Frost (1906);
68~= Sanford (1927);
69~= Wallerstein (1972); 
70~= Coulson (1983); 
71~= Evans \& Lyons (1994);
72~= Gieren (1989); 
73~= Harper (1934); 
74~= Coulson et~al. (1985); 
75~= Gieren (1985); 
76~= Gieren (1981a); 
77~= Jacobsen (1970);
78~= Breger (1967);
79~= Sanford (1951);
80~= Imbert (1996); 
81~= Coulson \& Caldwell (1985b); 
82~= Eggen (1983); 
83~= Evans \& Sugars (1997); 
84~= Hipparcos Epoch Photometry;
85~= Babel et~al. (1989); 
86~= Kimeswenger et~al. (2004); 
87~= Albrow \& Cottrell (1996); 
88~= Jacobsen et~al. (1984); 
89~= Jacobsen (1974); 
90~= Henden (1980); 
91~= Evans \& Lyons (1992); 
92~= Abt \& Levy (1978); 
93~= Duncan (1922);
94~= ten Bruggencate (1930); 
95~= Evans \& Welch (1993); 
96~= Gieren (1989);
97~= Sugars \& Evans (1996); 
98~~= Fernley et~al. (1989);
99~= Dean (1977); 
100~= Dean (1981); 
101~= Duncan (1932); 
102~= Slipher (1904);
103~= Moore (1909); 
104~= Feast et~al. (2008);
105~= Metzger et~al. (1993);
106~= Evans (1988);
107~= Maddrill (1906);
108~= Abt (1973);
109~= Imbert (1985);
110~= Joy (1952).
\par
Note: 1~= The number indicates the dataset not used in, respectively, $V$, $K$, RV.
\par
\label{TAB-DATA} 
\end{table}